ORIGINAL REPORT

CHANGES IN PERCEIVED IMPACT OF STROKE ON EVERYDAY LIFE OVER FIVE YEARS IN A REHABILITATION SAMPLE THAT RECEIVED AN ACTIVITY OF DAILY LIVING INTERVENTION: A FOLLOW-UP STUDY

Mandana FALLAHPOUR, PHD¹, Gunilla ERIKSSON, PHD¹,² and Susanne GUIDETTI, PHD¹ From the ¹Department of Neurobiology, Care Sciences and Society, Division of Occupational Therapy, Karolinska Institutet, Stockholm and ²Department of Neuroscience, Rehabilitation Medicine, Uppsala University, Uppsala, Sweden

Objective: To compare changes in the perceived impact of stroke on everyday life over time in a rehabilitation sample that received a client-centred activities of daily living (CADL) intervention or usual ADL (UADL) intervention.

Design: Longitudinal follow-up of a randomized controlled trial.

Methods: A total of 145 persons with stroke were assigned into CADL or UADL. Groups were assessed using the Stroke Impact Scale (SIS) at 3 months, 12 months and 5 years post-intervention. Changes in SIS domain scores over time were compared within and between groups.

Results: Changes in the impact of stroke over time were not related to which intervention the groups received. There were no significant differences in the SIS domains or stroke recovery between groups at the 3-month, 12-month and 5-year follow-ups. Despite an increased impact of stroke over time in some domains in both groups, both groups perceived a decreased impact of stroke in the Participation domain at 12 months. Perceived participation was sustained at the same level at 12 months as at 5 years in both groups. Conclusion: These findings stress the importance of access to follow-up rehabilitation interventions 1-year post-stroke to enable participation in daily activities. Such follow-up and enablement would support the use of self-management strategies in the performance of persons' valued activities, which might be difficult to perform, due to, for example, impact on hand function or mobility. The results of this study emphasize the importance of prioritizing participation in activities that are meaningful from a personal perspective.

Key words: stroke; participation; everyday life; daily activities; rehabilitation; occupational therapy; follow-up studies; longitudinal studies.

Accepted April 20, 2022; Epub ahead of print 18 May, 2022

J Rehabil Med 2022; 54: jrm00291

DOI: 10.2340/jrm.v54.1060

Correspondence address: Mandana Fallahpour, Karolinska Institutet, Department of Neurobiology, Care Sciences and Society, Division of Occupational Therapy, Fack 23200, SE-141 83 Huddinge, Stockholm, Sweden. E-mail: Mandana.Fallahpour@ki.se

LAY ABSTRACT

In this study, perceived impact of stroke on everyday life over time was compared among 145 persons with stroke who received the client-centred or usual activities of daily living intervention. They were followed up at 3, 12 months and 5-years post-stroke. No differences were found in the impact of stroke over time between groups. Despite increased impact of stroke (indicating more problems) over time in some areas, the groups perceived decreased impact of stroke (less problems) in their participation at 12 months. Perceived participation was sustained at the same level at 12 months as at 5 years in both groups. Findings stress the importance of access to followup rehabilitation interventions one-year post-stroke to enable participation in daily activities, and to support the use of self-management strategies to facilitate participation in valued and meaningful activities that might be difficult due to person's functional limitations.

Troke is one of the common leading causes of disability worldwide (1), and impacts individuals' functioning and participation in everyday life (2–5). The long-term consequences of stroke might also affect individuals' life satisfaction (5–7). Previous research shows that a large number of individuals (> 50%) in a hospital-based sample 1-year post-stroke still experienced restrictions in participation in daily activities (8). Rehabilitation interventions are mostly recommended in the first 3 months post-stroke, due to the importance of this time-period in plasticityenhancing mechanisms and significant stroke recovery (9). Previous studies, however, suggest that, in 1 year or longer post-stroke, strength and hand function (10–12), activities of daily living (ADL), health-related quality of life (10, 13) and participation (12, 13) are still severely impacted. Moreover, the needs for rehabilitation 1 year post-stroke are not always considered to be met (14), particularly among persons who have had a moderate-to-severe stroke (8, 15). This indicates an essential need for developing effective rehabilitation interventions targeting participation in everyday life and exploring individuals' needs over time. Moreover, sustaining participation

is of great importance and priority. A few studies in various contexts have focused on the perceived impact of stroke at different time-points (12, 16–20) assessed using the Stroke Impact Scale (SIS 3.0). This scale includes 8 domains: Strength; Memory and thinking; Emotions; Communication; Mobility; Activities of daily living (ADL) and Instrumental ADL (IADL); Hand function; and Participation (21). Strength, Hand function, and Participation were the domains in SIS 3.0 that were most affected at 3 months post-stroke (12, 22). Furthermore, the ADL and IADL domains, and the Mobility domain, were also impacted at 3 months post-stroke (22). The Strength and Hand function domains (10–12, 19) and the Participation domain (12, 13, 19) were found to be the most affected at 1 year post-stroke (12, 19). Previous long-term follow-ups post-stroke mostly explored the impact of stroke up to 2 years (10, 23, 24). A follow-up study 6 years post-stroke (19) identified the most impacted SIS domains as: Participation, Strength, Hand function, and Stroke recovery. Those with moderate-to-severe stroke experienced a higher impact in all domains, except Hand function and Stroke recovery, indicating more problems in everyday life, compared with those with mild stroke. A recent follow-up study (20) showed that the perceived consequences of stroke were more severe after 5 years compared with at 1 year. Strength, Emotion and Participation were the areas most affected, along with restrictions in social life and autonomy indoors. The perceived impact of stroke could therefore become more prominent with time, even for persons with mild-to-moderate stroke, highlighting the need for long-term support for persons with stroke. The level of dependency at discharge from hospital was found to be of importance in self-perceived impacts 5 years post-stroke in another follow-up study (25). For those who were functionally independent, non-modifiable factors, e.g. age, sex and type of stroke, were significant predictors of a better outcome. However, for the functionally dependent persons, modifiable factors, such as feeling depressed, cardiovascular risk factors, and recurrent stroke, were significant predictors of unfavourable outcome (25). Previous research has made a great effort in describing the consequences of stroke over time. However, there is still limited knowledge regarding how individuals who have received rehabilitation interventions perceive the longterm impacts of stroke on everyday life. Rehabilitation interventions are mostly focused on the first 3 months post-stroke. To be able to plan and deliver appropriate long-term rehabilitation interventions and follow-ups that are individually adapted (26), perceived changes over time post-stroke need to be explored among persons with stroke who are in need of rehabilitation. Changes after rehabilitation need to be compared among groups that have received different intervention programmes.

A client-centred ADL intervention (CADL) was developed (27) to be applied among persons with stroke (16) and further compared with usual ADL interventions (UADL) in phase 2 of the Life After Stroke study (LAS II) (16). The interventions were evaluated in a randomized controlled trial (RCT) and changes between groups were compared at 3, 6, 12 months, and 5 years after inclusion (16–18). No differences were found in the participation outcome between the 2 groups at the 3-month follow-up (16), 6-month and 1-year follow-ups (17) and 5-year follow-up (18). However, in the 3-month follow-up a significant difference was found in the SIS domain of Emotion in favour of the CADL intervention (16). The 1-year follow-up, including the whole sample who received CADL and UADL, showed no statistically significant differences between the groups regarding perceived Participation (17). Nevertheless, a positive trend was observed regarding a clinically meaningful positive change in the CADL group on perceived Participation measured by SIS (domain 8).

The 5-year follow-up study showed that perceived Participation (SIS domain 8) improved over time for all participants, regardless of whether they received the CADL or UADL intervention (18). The completed RCT study, comparing 2 groups that received different rehabilitation interventions, provided the opportunity to study the perceived impact of stroke over time. However, in an ageing group of people, many incidents in everyday life can occur in such a long period as 5 years. As such, one should be aware that the type of rehabilitation received might have limited influence regarding the perceived impact of stroke.

Longitudinal studies that focus on the consequences of stroke are crucial to understand the impact of stroke on people's everyday life and how the impact of stroke might change over time. This knowledge is scarce. In our previous study the impact on the domain Participation was reported at 5 years poststroke (17). Understanding the impact people perceive on their everyday functioning, beyond participation, over the 5 years since stroke onset, is essential to further develop rehabilitation interventions that meet individual needs. The overall aim of this study was therefore to compare changes in the perceived impact of stroke on everyday life over time in a rehabilitation sample who have received either the CADL or UADL intervention.

METHODS

Design and participants

The study was designed as a longitudinal 5-year follow-up of a RCT. The methodology, inclusion criteria, recruitment procedure, and results of the original trial (16, 17, 28) and the 5-year follow-up study (18) are described fully in earlier publications. Briefly, in the original trial (16), 16 rehabilitation units in 3 Swedish county councils were included and randomized to provide the CADL intervention and UADL intervention as intervention and control groups, respectively. Persons with stroke admitted to these units between 6 October 2009 and 7 September 2011 were recruited as participants. In the 5-year follow-up study, the recruitment occurred between 2014 and 2016. All the participants from the original trial who were alive, verified by the Swedish population registry, were contacted first by telephone, and then by letter if there was no response from the person. Information about the study was provided and, in case the person was interested in participation, a date was booked for data collection in participant's home, and informed consent was given. In this follow-up study 145 persons with stroke were recruited and gave their informed consent.

The mean duration of rehabilitation was 40 days (range 7–120 days) for these persons who had been admitted to the participating units for 1 year. Baseline characteristics of the participants in the 2 groups (CADL or UADL) are shown separately in Table I.

The original trial (ClinicalTrials.gov NCT01417585) and the 5-year follow-up received ethics approval from the Regional Ethical Review Board in Stockholm with the reference numbers 2009/727-31/1 and 2014/996-32.

Data collection

Persons with stroke who met the inclusion criteria and agreed to take part in the RCT after giving their informed consent were assessed over time, i.e. at 3 months, 12 months and 5 years, by the same data collectors, who were blinded to the group allocation. More details about the data collection procedure are provided in previous publications (16–18). In this study the data collected from the following outcome measures are used:

the Stroke Impact Scale 3.0 (SIS) (21, 29) was used to assess the perceived impact of stroke on 8 various domains of Strength, Memory and thinking, Emotions, Communication, Mobility, ADL/IADL, Hand function, and Participation. The SIS comprises 59 items with scores ranging from 0 to 100, where a higher score reflects a smaller impact. Furthermore, the SIS includes 1 item, Stroke recovery, which was assessed with a visual

- analogue scale, ranging from 0 (no recovery) to 100 (full recovery). For those participants who were not able to respond to the questions, a proxy version (30) was used.
- the Mini-Mental State Examination (MMSE) (31) was used to assess cognitive function, ranging from 0 to 30 scores.
- the Barthel Index (BI) (32) was used to assess independence in ADL. The BI scores, ranging from 0 to 100, were also used to determine stroke severity (33) and assign the participants into 3 groups of mild stroke (50–100), moderate stroke (15–49), and severe stroke (less than 14).

Data analysis

Descriptive statistics were used to present the characteristics of the participants at baseline.

Baseline variables were compared between the 2 groups (CADL and UADL) using independent *t*-test for continuous variables and χ^2 for categorical variables. Linear mixed models of analysis of variance (ANOVA) were used to examine differences in the change in impact of stroke, as verified by SIS recovery and SIS domain scores 1–8. After inclusion in the RCT, this

Table I. Baseline characteristics of the participants in client-centred activities of daily living (CADL) intervention or usual ADL (UADL) intervention groups (n = 145)

Characteristics	CADL	UADL
n	71	74
Age, years		
Mean (SD)	71 (9)	68 (9)
Sex, n (%)		
Male	41 (58)	48 (65)
Female	30 (42)	26 (35)
Education, n (%)		
Elementary school	32 (45)	30 (41)
High school	17 (24)	23 (31)
University	22 (31)	21 (28)
Living status, n (%)		
Living alone	44 (62)	49 (66)
Living with someone	27 (38)	25 (34)
Type of stroke, n (%)		
Infarct	52 (73)	53 (72)
Haemorrhagic	11 (15)	12 (16)
Unspecified	8 (11)	9 (12)
Hemisphere lesion, n (%)		
Left	33 (46)	32 (43)
Right	36 (51)	39 (53)
Unspecified	2 (3)	3 (4)
Stroke severity, n (%)		
Mild (BI = $50-100$)	55 (78)	70 (95)
Moderate (BI = 15-49)	15 (21)	4 (5)
Severe (BI < 15)	1(1)	0 (0)
Independence in ADL (Barthel Index = 0-100)		
Mean (SD)	66 (23)	78 (18)
Cognitive function (MMSE = $0-30$)		
Mean (SD)	25.4 (5.8)*	26.5 (4.4)*
Stroke Impact: recovery (SIS = 0-100)		
Mean (SD)	33 (22)	41 (23)

CADL: client-centred activities of daily living; UADL: usual activities of daily living; BI: Barthel index; MMSE: Mini-Mental State Examination; SIS: Stroke Impact Scale.

^{*}Missing data = 2.

change in impact was measured within and between each group of CADL and UADL over different time-points of 3 months, 12 months and 5 years. Pair-wise comparisons were adjusted for multiple comparisons using Bonferroni correction to further explore the changes between different time-points. Moreover, the measures of SIS recovery and SIS domains 1–8 were compared for each time-point between the CADL and UADL groups using analysis of covariance (AN-COVA). Covariates included in the statistical analyses were age, sex, stroke severity (BI score), independence in ADL (BI score) and stroke recovery (SIS score) at baseline.

A *p*-value < 0.05 was set as statistically significant. The analyses were conducted using the SPSS program version 27.0 (Statistical Package for the Social Sciences, Armonk, NY: IBM Corp).

RESULTS

Baseline data for participants in the 5-year follow-up are shown in Table I. The results of Pearson's χ^2 analysis to compare categorical variables at baseline did not demonstrate any significant differences between frequency of sex (p = 0.4; $\chi^2 = 0.78$), education (p = 0.93; $\chi^2 = 0.64$) and living status (p = 0.61; $\chi^2 = 0.28$) in the 2 groups. The only exception in this comparison was stroke severity (p = 0.004; $\chi^2 = 9.11$), which showed that the number of persons with mild stroke was higher in the UADL group than in the CADL group. The results of the independent t-tests to compare the mean values of continuous variables in 2 groups

at baseline did not show any significant differences for age (p = 0.08) and cognitive function verified by MMSE score (p = 0.2). However, there were significant differences in the mean BI total score (p = 0.001) and SIS recovery (p = 0.04) scores between the 2 groups at baseline. These results showed a higher mean value for independence in ADL and stroke recovery in the UADL group compared with the CADL group (see Table I).

The findings of the ANOVA mixed models comparing change in the impact of stroke as verified by SIS domains 1–8 and recovery scores between 3 months, 12 months, and 5 years post-stroke within each group of CADL and UADL are shown in Tables II–III.

In the pairwise comparisons of the mean values between different time-points in the CADL group, statistically significant changes were found in the domains of: (i) Communication between 3 months and 5 years, and 12 months and 5 years post-stroke with an increased impact of stroke; (ii) Mobility between 12 months and 5 years post-stroke with an increased impact of stroke; (iii) Hand function between 12 months and 5 years with an increased impact of stroke, and, finally; (iv) Participation between 3 and 12 months, and 3 months to 5 years post-stroke with a decreased impact of stroke in both time-periods (see Table II).

In the pairwise comparisons of the mean values between different time-points in the UADL group, statistically significant changes were found in domains of: (*i*) Emotion between 3 months and 5 years post-stroke with a decreased impact of stroke; (*ii*) Mobility between 12 months and 5 years with an increased impact of stroke; (*iii*) Hand function between 3 and 12 months post-stroke

Table II. Comparing changes in the impact of stroke between different time-points post-stroke in the client-centred activities of daily living (CADL) group using linear mixed models of analysis of variance (ANOVA) (n = 71)

	3 months (T1)	12 months (T2)	5 years (T3)	Changes within CADL group	1	Pairwise comparisons of mean values between different time-points ($ ho ext{-value}$)			
SIS domains	CADL	CADL	CADL	<i>p</i> -value	-	T1-T2	T1-T3	T2-T3	
Mean (SD)					F (df, error df)	(3-12 months)	(3 months - 5 years)	(12 months - 5 years)	
SIS 1	60.2	57.8	57.9	0.520	F (2, 130) = 0.658	0.864	1.000	1.000	
Strength	(21.6)	(22.8)	(24.0)						
SIS 2	85.3	84.6	80.5	0.022	F(2, 132) = 3.948	1.000	0.113	0.068	
Memory/thinking	(16.6)	(15.6)	(20.6)						
SIS 3	74.6	73.1	74.2	0.611	F(2, 130) = 0.494	0.857	1.000	1.000	
Emotion	(15.3)	(14.2)	(17.2)						
SIS 4	84.9	86.0	80.2	0.005	F(2, 130) = 5.513	1.000	0.035 (II)	0.009 (II)	
Communication	(20.8)	(15.2)	(20.9)						
SIS 5	73.4	75.8	72.9	0.298	F(2, 130) = 1.222	0.307	1.000	0.455	
ADL/IADL	(18.4)	(17.1)	(22.9)						
SIS 6	76.5	77.1	69.8	0.002	F(2, 130) = 6.323	1.000	0.058	0.003 (II)	
Mobility	(21.9)	(20.2)	(25.8)						
SIS 7	49.5	52.5	45.3	0.027	F(2, 128) = 3.719	0.715	0.359	0.033 (II)	
Hand function	(33.9)	(33.7)	(36.5)						
SIS 8	56.8	66.5	68.5	< 0.001	F (2, 130) = 11.522	< 0.001 (DI)	< 0.001 (DI)	1.000	
Participation	(20.9)	(20.5)	(20.8)						
SIS Recovery	53.8	57.7	57.7	0.241	F(2, 128) = 1.438	0.359	0.515	1.000	
	(21.7)	(22.9)	(24.9)						

T: time post-stroke; DI: decreased impact of stroke; II: increased impact of stroke

Higher scores present/reflect smaller impact of stroke.

Bold numbers present significant differences as verified by p-value less than 0.05.

Table III. Comparing changes in the impact of stroke between different time-points post-stroke in the usual activities of daily living (UADL) group using linear mixed models of analysis of variance (ANOVA) (n = 74)

	3 months (T1) UADL	12 months (T2)	5 years (T3)	Changes within UADL group		Pairwise comparisons of mean values between different time-points (p value)			
SIS items Mean (SD)			UADL	<i>p</i> -value	F (df, error df)	T1-T2 (3-12 months)	T1-T3 (3 months - 5 years)	T2-T3 (12 months - 5 years)	
SIS 1	62.4	64.5	67.5	0.056	F (2, 136) = 2.952	0.865	0.093	0.437	
Strength	(24.4)	(24.8)	(26.8)						
SIS 2	84.6	86.1	83.6	0.269	F(2, 136) = 1.326	1.000	1.000	0.144	
Memory/thinking	(16.2)	(15.7)	(15.9)						
SIS 3	73.6	76.3	78.7	0.009	F(2, 136) = 4.910	0.204	0.021 (DI)	0.400	
Emotion	(14.5)	(13.9)	(16.3)						
SIS 4	88.1	88.4	85.8	0127	F(2, 136) = 2.083	1.000	0.467	0.177	
Communication	(14.9)	(11.9)	(17.4)						
SIS 5	78.9	82.3	79.9	0.072	F(2, 134) = 2.683	0.09	1.000	0.487	
ADL/IADL	(19.6)	(17.8)	(22.8)						
SIS 6	82.2	82.9	78.3	0.012	F(2, 134) = 4.567	1.000	0.102	0.014 (II)	
Mobility	(19.9)	(17.6)	(21.9)						
SIS 7	62.8	67.8	64.2	0.044	F(2, 132) = 3.208	0.037 (DI)	1.000	0.202	
Hand function	(36.5)	(34.0)	(35.1)						
SIS 8	64.7	71.6	75.5	< 0.001	F(2, 140) = 9.839	0.007 (DI)	0.001 (DI)	0.318	
Participation	(21.4)	(20.5)	(21.0)						
SIS Recovery	59.9	68.3	64.5	0.005	F (2, 138) = 5.484	0.001 (DI)	0.351	0.383	
	(23.2)	(22.0)	(27.2)						

T: time post-stroke; DI: decreased impact of stroke; II: increased impact of stroke.

Bold numbers present significant differences as verified by p-value less than 0.05.

with a decreased impact of stroke; (*iv*) Participation between 3 and 12 months, and 3 months and 5 years post-stroke with a decreased impact of stroke in both time-periods, and, finally; (*v*) Recovery between 3 and 12 months post-stroke with a decreased impact of stroke (see Table III).

Table IV presents a comparison of the changes in the impact of stroke regarding measures of SIS domains 1–8 and stroke recovery between the 2 groups of CADL and UADL over the 5 years post-stroke using the analysis of covariance (ANCOVA). The results show that these changes were not related

to which intervention participants received in the CADL or UADL groups.

DISCUSSION

This longitudinal study aimed to compare changes in the perceived impact of stroke in a rehabilitation sample that received CADL and UADL intervention between 3 different time-points of 3 months, 12 months, and 5 years post-intervention. This study made it possible to explore the perceived impact of stroke on people in different time-points over a period of 5 years, which was

Table IV. Comparing changes in the impact of stroke between 2 groups of client-centred activities of daily living (CADL) intervention or usual ADL (UADL) over time post-stroke using analysis of covariance (ANCOVA) (n = 145)

SIS domains Mean (SD)	3 months			12 months			5 years		
	CADL	UADL	p-value	CADL	UADL	<i>p</i> -value	CADL	UADL	<i>p</i> -value
SIS 1	60.2	62.4	0.930	57.8	64.5	0.903	57.9	67.5	0.578
Strength	(21.6)	(24.4)		(22.8)	(24.8)		(24.0)	(26.8)	
SIS 2	85.3	84.6	0.580	84.6	86.1	0.894	80.5	83.6	0.744
Memory/thinking	(16.6)	(16.2)		(15.6)	(15.7)		(20.6)	(15.9)	
SIS 3	74.6	73.6	0.213	73.1	76.3	0.738	74.2	78.7	0.869
Emotion	(15.3)	(14.5)		(14.2)	(13.9)		(17.2)	(16.3)	
SIS 4	84.9	88.1	0.606	86.0	88.4	0.683	80.2	85.8	0.460
Communication	(20.8)	(14.9)		(15.2)	(11.9)		(20.9)	(17.4)	
SIS 5	73.4	78.9	0.157	75.8	82.3	0.600	72.9	79.9	0.318
ADL/IADL	(18.4)	(19.6)		(17.1)	(17.8)		(22.9)	(22.8)	
SIS 6	76.5	82.2	0.151	77.1	82.9	0.772	69.8	78.3	0.355
Mobility	(21.9)	(19.9)		(20.2)	(17.6)		(25.8)	(21.9)	
SIS 7	49.5	62.8	0.445	52.5	67.8	0.683	45.3	64.2	0.858
Hand function	(33.9)	(36.5)		(33.7)	(34.0)		(36.5)	(35.1)	
SIS 8	56.8	64.7	0.379	66.5	71.6	0.959	68.5	75.5	0.503
Participation	(20.9)	(21.4)		(20.5)	(20.5)		(20.8)	(21.0)	
SIS recovery	53.8	59.9	0.437	57.7	68.3	0.548	57.7	64.5	0.699
	(21.7)	(23.2)		(22.9)	(22.0)		(24.9)	(27.2)	

 $SIS: Stroke\ Impact\ Scale;\ CADL:\ client-centred\ activities\ of\ daily\ living;\ UADL:\ usual\ activities\ of\ daily\ living.$

Higher scores present/reflect smaller impact of stroke.

a unique possibility and, to our knowledge, not studied previously. However, when we conducted the study, we were aware that many things could happen that might affect the lives of people at older ages over the years. On this basis, we reasoned that the type of rehabilitation intervention received earlier might not have the most significant effect on everyday life at long-term follow-up. The study results aligned with this reasoning showing that changes in the impact of stroke (SIS) for participants at 3-month, 12-month and 5-year follow-ups were not related to which intervention they received, as no significant differences were found between the CADL and UADL groups in the SIS domains. These results are in line with previous follow-ups carried out at 3 (16) and 12 months (17).

The CADL group demonstrated a significant decreased impact of stroke in the domain of Participation between follow-ups at 3 months and 12 months, and 3 months to 5 years. This indicates that they perceived that their participation in everyday life was enhanced. Most clinically meaningful changes, both negative and positive, were found in the Participation domain in a previous study (12), indicating that this domain is sensitive at follow-ups. However, there was no difference in Participation between 12 months and 5 years, indicating that perceived participation was enhanced in the first year and maintained 4 years later. Interestingly, despite less impact on Participation, the CADL group had a significantly increased impact of stroke (more problems) on Communication, Mobility, and Hand function at the 12-month and 5-year follow-ups, which might be considered to be reflected in the participation score sustained at the decreased 12-month level. This is in line with the results of a recent study that found no significant differences between the perceived impact of stroke on participation between 1 and 6 years post-stroke, even if the impact of stroke significantly increased in Strength, ADL/IADL, Mobility and Hand function domains between these 2 time-points (19). Another previous study showed that an increased impact of stroke was found in Strength, Emotion, Communication, Mobility and even Participation domains after 5 years. The UADL group had similar perceptions of their Participation, even if a significant increased impact of stroke was found on their Mobility between follow-ups at 12 months and 5 years. Previous research (20), however, showed that the perceived impact of stroke was more severe after 5 years compared with at the first years. Strength, Emotion and even Participation were the areas most affected. One possible reason for the finding in the current study might be that the participants in both groups had found a way to deal with the consequences implied by stroke through management strategies they developed over time. Despite slightly lower body function after a stroke, they

could solve problems in their doings to improve their participation in everyday life. It might also be that the participants had adapted to a declined activity repertoire at an older age. Even if the stroke had affected their ability to participate in their usual activities, it can be assumed that they could participate in activities that, to them, were both meaningful and helped provide purpose in life (6). In line with that, perhaps better perceived participation resulted from participants stopping activities that did not work for them and, rather, continuing activities that were meaningful to them. These findings emphasize that the perceived impact of stroke could become more prominent over time. However, Emotion, Participation and Stroke recovery could be enabled even if other domains show restrictions and deterioration. Future research should therefore focus on identifying how to better enable participation in meaningful activities considering individuals' personal strategies and perspectives.

It is notable that the participants estimated the impact of their stroke at approximately the same level after 5 years as at the 3-month follow-up in most of the SIS domains. Thus, the participants are approximately "in the same place" regarding their perceived impact of stroke at 5 years as at shortly after their stroke, but, despite this, they perceived that their participation had enhanced. These findings are clinically important in stroke rehabilitation and stress the importance of prioritizing the enablement of participation in everyday life, rather than focusing mostly on body functions and impairments. From an occupational perspective, these findings are also important. Prioritizing participation in activities that are meaningful from a personal perspective is the basis for providing a client-centred rehabilitation (34). Another clinical implication of the findings is that it is important to consider and evaluate the need for rehabilitation over time among persons with stroke and their need to have longer contact and the opportunity to be offered a new rehabilitation period to receive the support they need.

A decreased impact of stroke over time (indicating fewer problems) in the domains of Emotions, Hand function, and Stroke recovery was found in the UADL group. These results were not found in the CADL group (see Tables II and III). One possible reason for these differences between the 2 groups could relate to the lower severity of stroke, higher independence in ADL, and higher perceived Stroke recovery at baseline in the UADL group. Given that no group differences were detected in previous follow-ups (16–18), a possible explanation is that the interventions in the experimental and control groups might have been too similar. As discussed previously (18), it is plausible that the UADL intervention might have comprised client-centred components due to occupational therapists' long history and experiences of applying a client-centred approach

in clinical practice (35). Previous qualitative research exploring occupational therapists' experiences in the context of providing a client-centred ADL intervention identifies important elements/characteristics, such as responsibility and ownership of the rehabilitation process among persons with stroke (36). These characteristics could also be found in usual occupational therapy interventions, making it difficult to differentiate between the effects of the interventions provided to the CADL and UADL groups in this study.

Study limitations

It is important to reflect on the generalizability of the results of this study to the population of people who have had stroke worldwide. In this study 78% of the CADL group and 95% of the UADL group in the sample had had a mild stroke. Moreover, there was only 1 person with a severe stroke in the CADL group and no subjects with a severe stroke in the UADL group. The significantly higher number of persons with a mild stroke in the UADL group than the CADL group might have affected the results. Furthermore, since the majority of the total sample (n = 125) were diagnosed with a mild stroke, the generalizability of the results to a broader group of people with stroke, and especially to those with severe stroke with serious problems in ADL, might be limited. Future studies are therefore needed to explore the changes in impact of stroke in this group. Another reflection on the generalizability of the results might be that this study set out to compare changes in the perceived impact of stroke over time in a rehabilitation sample who have received either the CADL or UADL intervention, and thus may provide limited knowledge that can be generalized to a wider stroke population. Another limitation in the study is the lack of power, since the drop-out analysis showed that more vulnerable persons with stroke who were in a poorer position had dropped out of the study. This situation, however, was evenly distributed across the 2 groups.

Addressing the impact of stroke over 5 years is challenging. One might assume that, in an ageing group of people, many incidents in everyday life can occur in such a long period as 5 years, and that the type of rehabilitation received might have limited influence regarding the perceived impact of stroke. It is therefore reasonable to consider the normal ageing process as a factor. This may be reflected in several domains of the SIS, such as mobility, memory and thinking, and hand function, where an impact of ageing rather than stroke alone can be expected.

In conclusion, despite the increased impact of stroke in some domains in the 2 groups over time, the participants in both groups perceived decreased impact of stroke on the domain of Participation at 12 months. In both groups the perceived participation was sustained at the same level at 12 months as at 5 years. This result is clinically important, as there seems to be a need to prioritize the enablement of participation in daily activities in the long-term, after 12 months post-stroke among persons with stroke, rather than focusing mostly on body functions and impairments. There might be a need for access to follow-up rehabilitation interventions, even years post-stroke, to support the use of self-management strategies in the performance of persons' valued activities in everyday life. The results of this study emphasize the importance of prioritizing participation in activities that are meaningful from a personal perspective, in order to provide client-centred rehabilitation over a longer period of time.

ACKNOWLEDGEMENTS

The authors are grateful to the participants in the study for their valuable contribution. The research project was financed by the Swedish Research Council (VR), the Swedish Stroke Association, and the Uppsala-Örebro Regional Research Council.

The authors have no conflicts of interest to declare.

REFERENCES

- GBD 2016 Stroke Collaborators. Global, regional, and national burden of stroke, 1990–2016: a systematic analysis for the Global Burden of Disease Study 2016. Lancet Neurol 2019; 18: 439–458.
- Eriksson G, Tham K, Borg J. Occupational gaps in everyday life 1–4 years after acquired brain injury. J Rehabil Med 2006; 38: 159–165.
- Bergstrom AL, Guidetti S, Tistad M, Tham K, von Koch L, Eriksson G. Perceived occupational gaps one year after stroke: an explorative study. J Rehabil Med 2012; 44: 36–42.
- Fallahpour M, Tham K, Joghataei MT, Jonsson H. Perceived participation and autonomy: aspects of functioning and contextual factors predicting participation after stroke. J Rehabil Med 2011; 43: 388–397.
- Fallahpour M, Tham K, Joghataei MT, Eriksson G, Jonsson H. Occupational gaps in everyday life after stroke and the relation to functioning and perceived life satisfaction. OTJR: Occupation, Participation and Health 2011; 31: 200–208.
- Bergstrom A, Guidetti S, Tham K, Eriksson G. Association between satisfaction and participation in everyday occupations after stroke. Scand J Occup Ther 2017; 24: 339–348.
- Eriksson G, Kottorp A, Borg J, Tham K. Relationship between occupational gaps in everyday life, depressive mood and life satisfaction after acquired brain injury. J Rehabil Med 2009: 41: 187–194.
- Riks-Stroke. [One year after stroke Report from the Swedish Stroke Register presented in december 2015; One year follow-up 2014] [Ett år efter stroke - Rapport från Riskstroke utgiven december 2015; 1-årsuppföljning 2014] [accessed 2022 Feb] [Available from: https://www. riksstroke.org/wp-content/uploads/2015/12/Riksstroke_1-%C3%A5rsuppf%C3%B6ljning_LR_13_14.pdf (in Swedish).
- Grefkes C, Fink GR. Recovery from stroke: current concepts and future perspectives. Neurol Res Pract 2020; 2: 17.

- Carod-Artal FJ, Coral LF, Trizotto DS, Moreira CM. The stroke impact scale 3.0: evaluation of acceptability, reliability, and validity of the Brazilian version. Stroke 2008; 39: 2477–2484.
- Hartman-Maeir A, Eliad Y, Kizoni R, Nahaloni I, Kelberman H, Katz N. Evaluation of a long-term community based rehabilitation program for adult stroke survivors. NeuroRehabilitation 2007; 22: 295–301.
- Guidetti S, Ytterberg C, Ekstam L, Johansson U, Eriksson G. Changes in the impact of stroke between 3 and 12 months post-stroke, assessed with the Stroke Impact Scale. J Rehabil Med 2014; 46: 963–968.
- Muren MA, Hutler M, Hooper J. Functional capacity and health-related quality of life in individuals post stroke. Top Stroke Rehabil 2008; 15: 51–58.
- 14. Tistad M, von Koch L, Sjostrand C, Tham K, Ytterberg C. What aspects of rehabilitation provision contribute to self-reported met needs for rehabilitation one year after stroke--amount, place, operator or timing? Health Expect 2013; 16: e24-35.
- Tistad M, Tham K, von Koch L, Ytterberg C. Unfulfilled rehabilitation needs and dissatisfaction with care 12 months after a stroke: an explorative observational study. BMC Neurol 2012; 12: 40.
- Bertilsson AS, Ranner M, von Koch L, Eriksson G, Johansson U, Ytterberg C, et al. A client-centred ADL intervention: three-month follow-up of a randomized controlled trial. Scand J Occup Ther 2014; 21: 377–391.
- Guidetti S, Ranner M, Tham K, Andersson M, Ytterberg C, von Koch L. A "client-centred activities of daily living" intervention for persons with stroke: One-year follow-up of a randomized controlled trial. J Rehabil Med 2015; 47: 605–611.
- 18. Hedman A, Eriksson G, von Koch L, Guidetti S. Five-year follow-up of a cluster-randomized controlled trial of a client-centred activities of daily living intervention for people with stroke. Clin Rehabil 2019; 33: 262–276.
- 19. Ytterberg C, Dyback M, Bergstrom A, Guidetti S, Eriksson G. Perceived impact of stroke six years after onset, and changes in impact between one and six years. J Rehabil Med 2017; 49: 637–643.
- Skoglund E, Westerlind E, Persson HC, Sunnerhagen KS. Self-perceived impact of stroke: A longitudinal comparison between one and five years post-stroke. J Rehabil Med. 2019; 51(9): 660–664.
- Duncan PW, Bode RK, Min Lai S, Perera S, Glycine antagonist in neuroprotection Americans I. Rasch analysis of a new stroke-specific outcome scale: the Stroke Impact Scale. Arch Phys Med Rehabil 2003; 84: 950–963.

- 22. Lai SM, Studenski S, Duncan PW, Perera S. Persisting consequences of stroke measured by the Stroke Impact Scale. Stroke 2002; 33: 1840–1844.
- 23. Sturm JW, Dewey HM, Donnan GA, Macdonell RA, McNeil JJ, Thrift AG. Handicap after stroke: how does it relate to disability, perception of recovery, and stroke subtype?: the north North East Melbourne Stroke Incidence Study (NEMESIS). Stroke 2002; 33: 762–768.
- 24. Mackenzie AE, Chang AM. Predictors of quality of life following stroke. Disabil Rehabil 2002; 24: 259–265.
- 25. Westerlind E, Horsell D, Persson HC. Different predictors after stroke depending on functional dependency at discharge: a 5-year follow up study. BMC Neurol 2020; 20: 263.
- Guidetti S, Asaba E, Tham K. Meaning of context in recapturing self-care after stroke or spinal cord injury. Am J Occup Ther 2009: 63: 323–332.
- Guidetti S, Eriksson G, von Koch L, Johansson U, Tham K. Activities in daily living: the development of a new client-centred ADL intervention for persons with stroke. Scand J Occup Ther 2020; 29: 104–115.
- 28. Flink M, Bertilsson AS, Johansson U, Guidetti S, Tham K, von Koch L. Training in client-centeredness enhances occupational therapist documentation on goal setting and client participation in goal setting in the medical records of people with stroke. Clin Rehabil 2016; 30: 1200–1210.
- 29. Duncan PW, Wallace D, Lai SM, Johnson D, Embretson S, Laster LJ. The stroke impact scale version 2.0. Evaluation of reliability, validity, and sensitivity to change. Stroke 1999; 30: 2131–2140.
- Duncan PW, Lai SM, Tyler D, Perera S, Reker DM, Studenski S. Evaluation of proxy responses to the Stroke Impact Scale. Stroke 2002; 33 (11): 2593–2599.
- Folstein MF, Folstein SE, McHugh PR. "Mini-mental state". A practical method for grading the cognitive state of patients for the clinician. J Psychiatr Res 1975; 12: 189–198.
- 32. Mahoney FI, Barthel DW. Functional Evaluation: The Barthel Index. Md State Med J 1965; 14: 61–65.
- 33. Govan L, Langhorne P, Weir CJ. Categorizing stroke prognosis using different stroke scales. Stroke 2009; 40: 3396–3399.
- Kielhofner G. A model of human occupation: theory and application. 3rd edn. Baltimore: Williams and Wilkins: 2002.
- Hammell KR. Client-centred practice in occupational therapy: critical reflections. Scand J Occup Ther 2013; 20: 174–181.
- Ranner M, von Koch L, Guidetti S, Tham K. Client-centred ADL intervention after stroke: occupational therapists' experiences. Scand J Occup Ther 2016; 23: 81–90.